
-----------------------------------------------------------------------------
                Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
        Fri Oct 10 16:43:09 GMT 2008
Year: 08  Doy: 284
Observer: yasukawa
WEATHER COMMENT: Fri Oct 10 16:44:54 GMT 2008
In thick cirrus overcast, diffuse to no shadows at this time,
S wind 15 mph, temp 42 F.
___end___
A NEW TAPE HAS BEEN PUT INTO KAIEE DLT DRIVE, Fri Oct 10 16:45:44 GMT 2008
PSPT COMMENT: Fri Oct 10 16:59:09 GMT 2008
Starting up.
___end___
Fri Oct 10 16:59:40 GMT 2008    CHIP     Start Patrol
Fri Oct 10 16:59:45 GMT 2008    PICS     Start Patrol
PSPT COMMENT: Fri Oct 10 17:09:30 GMT 2008
Observing
___end___
**PSPT PROBLEM**: Fri Oct 10 17:20:33 GMT 2008
Image off to one side.  Aborted obs, recentering image, restarting obs.
___end___
**PSPT PROBLEM**: Fri Oct 10 17:38:25 GMT 2008
Image still drifting.  Loosened and retightened RA and dec clutches.
Recentering 
___end___
Fri Oct 10 18:00:19 GMT 2008    PICS     Flat
Fri Oct 10 18:02:45 GMT 2008    PICS     End Flat
Fri Oct 10 18:02:51 GMT 2008    PICS     ReStart Patrol
Fri Oct 10 18:02:53 GMT 2008    CHIP     LSD
Fri Oct 10 18:04:38 GMT 2008    CHIP     End LSD
Fri Oct 10 18:04:44 GMT 2008    CHIP     BiasLSD
Fri Oct 10 18:05:36 GMT 2008    CHIP     End BiasLSD
Fri Oct 10 18:05:42 GMT 2008    CHIP     Bias
Fri Oct 10 18:06:32 GMT 2008    CHIP     End Bias
Fri Oct 10 18:06:37 GMT 2008    CHIP     ReStart Patrol
Fri Oct 10 18:23:27 GMT 2008    CHIP     End Patrol
WEATHER COMMENT: Fri Oct 10 18:23:35 GMT 2008
Cirrus overcast very thick, standing down with obs for now.
___end___
Fri Oct 10 18:24:14 GMT 2008    PICS     End Patrol
Fri Oct 10 18:53:14 GMT 2008    CHIP     Start Patrol
Fri Oct 10 18:53:19 GMT 2008    PICS     Start Patrol
Fri Oct 10 19:24:29 GMT 2008    CHIP     End Patrol
Fri Oct 10 19:24:38 GMT 2008    PICS     End Patrol
COMMENT: Fri Oct 10 19:24:38 GMT 2008
Reconfiguring dome shutter.
___end___



Fri Oct 10 19:30:59 GMT 2008    CHIP     Start Patrol
Fri Oct 10 19:31:10 GMT 2008    PICS     Start Patrol
Fri Oct 10 21:10:52 GMT 2008    CHIP     End Patrol
Fri Oct 10 21:11:10 GMT 2008    PICS     End Patrol
WEATHER COMMENT: Fri Oct 10 21:11:08 GMT 2008
Clouds very thick again, standing down.
___end___
COMMENT: Fri Oct 10 21:12:24 GMT 2008
Just finished tour for 17 first-year astronomy grad students from
UH-IFA and 3 instructors lead by Klaus Hodapp.
___end___
Fri Oct 10 21:34:17 GMT 2008    CHIP     Start Patrol
Fri Oct 10 21:34:27 GMT 2008    PICS     Start Patrol
WEATHER COMMENT: Fri Oct 10 21:34:41 GMT 2008
Observing.
___end___
**PSPT PROBLEM**: Fri Oct 10 21:51:40 GMT 2008
suspect filter lost.  stopped and homed filter wheel
___end___
**PSPT PROBLEM**: Fri Oct 10 23:03:38 GMT 2008
C/k/r
___end___
Fri Oct 10 23:54:15 GMT 2008    CHIP     End Patrol
Fri Oct 10 23:54:19 GMT 2008    PICS     End Patrol
WEATHER COMMENT: Fri Oct 10 23:54:20 GMT 2008
Standing down, cloud cover is thick.
___end___
COMMENT: Sat Oct 11 00:17:47 GMT 2008
Reconfigured dome shutter.
___end___
NICE CHIP     IMAGE: 1934
NICE PICSDISC IMAGE: 1931
WEATHER COMMENT: Sat Oct 11 01:32:18 GMT 2008
Cirrostratus overcast very thick horizon to horizon, no shadows.
___end___
Sat Oct 11 01:32:58 GMT 2008 
        MkIV    
 


